
D0 Luminosity Monitor
- shutdown and commissioning work –

2007. 12.5 Yuji Enari 
At shutdown
- Replaced all scintillator due to radiation damage 

Update HV setting
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Replacement scintillator

• This is 2nd time
– 1st time was done 2006 shutdown.
– Delivered luminosity in RunIIb (1.7 fb-1) was larger 

than RunIIa (1.6 fb-1).  
• Replace all 48 counters.
• Not replace PMT.

– Put same PMT back same place.
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Successfully re-installed 

• Dismount August 14th.
• Reinstalled September 13th.      1 month task.
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First beam with new counters

First store: Oct 28 (Sun)
- Data with HV setting of before shutdown
Observe recovery on ADC/Lumi to same  
level at beginning of RunIIb.

- Update HV setting
Apply new HV setting mid of 1st store.    

- Update T0 (pass2 constant) between 1st

and 2nd store.
All channel look good and no problem

Normal operation has resumed since 3rd store.
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HV scan result

Default HV setting
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Threshold scan result

(mV)

Default setting : 30 mV
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Summary

• Replace all scintillator due to radiation damage
– PMTs are same, none  were replaced.

• Work has done successfully without any problem
– Light yields return to levels observed at start of RunIIb

• HV and Tzero was adjusted 1st and 2nd store. 
• Normal operation has resumed since 3rd store.
• Result of HV and Threshold scans indicate that we are  

operating on plateau.
• Very stable operation is on going.
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